Author's response to reviews

Title: RNA Profiling Reveals Familial Aggregation of Molecular Subtypes in non-BRCA1/2 Breast Cancer Families

Authors:

Martin J Larsen (martin.larsen@rsyd.dk)
Mads Thomassen (mads.thomassen@rsyd.dk)
Qihua Tan (gtan@health.sdu.dk)
Anne-Vibeke Lænkholm (anlae@regionsjaelland.dk)
Martin Bak (martin.bak@rsyd.dk)
Kristina P Sørensen (kristina.sorensen1@rsyd.dk)
Mette K Andersen (Mette.Klarskov.Andersen@regionh.dk)
Torben A Kruse (torben.kruse@rsyd.dk)
Anne-Marie Gerdes (anne-marie.gerdes@regionh.dk)

Version: 3 Date: 22 January 2014

Author's response to reviews: see over
Dear Editor

Thank you for the positive news, that our manuscript has been accepted by BMC Medical Genomics. In response to your request, we have added the following ethical statement to the Methods section of the revised manuscript:

“The study was carried out as a retrospective register study and in accordance with the Helsinki Declaration. The study has been approved The National Committee on Health Research Ethics of Denmark (S-V/F-200220142), waiving the requirement for informed consent for the study.”

We hope that by adding this, will fulfill the requirements. If the phrasing needs to be changed, please let us know.

Please find enclosed:
- Revised manuscript, including ethical statement

Yours sincerely,

Martin J. Larsen
martin.larsen@rsyd.dk